\documentclass[12pt]{article}
 
\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}
\newcommand{\answerspace}{\vfill}
%\setlength{\textwidth}{6in}
%\setlength{\oddsidemargin}{.30in}
\topmargin=-0.5in
\textheight=9in

 
\begin{document}
 
\textbf{Math 325: Differential Equations}
\hfill\makebox[2.5in][r]{Name:\hrulefill}
 
\textbf{Quiz 8}\quad \textsl{March 30, 2001}
 
\bigskip


You may  use your own calculator.
You may not use anything else.  You may not pass a calculator 
to another person.

\bigskip

Show all your work.  Erase or cross out any work you do not want graded.

\bigskip
\noindent
Find the eigenvalues and eigenfunctions of the problem
\begin{displaymath}
  y''+\lambda y=0,\qquad y'(0)=0,\ y'(L)=0.
\end{displaymath}
You may assume that all eigenvalues are real.

\end{document}

